Author’s response to reviews

Title: Impact of the donor-recipient gender matching graft survival from live donors donorrecipient gender matching on the graft survival from live donors

Authors:

Gholamhossein Naderi (Gh_naderi2000@yahoo.com)

Amin Azadfar (urc@tums.ac.ir)

Seyed Reza Yahyazadeh (r.yahyazadeh@gmail.com)

Fateme Khatami (f-khatami@alumnus.tums.ac.ir)

Seyed Mohammad Kazem Aghamir (mkaghamir@tums.ac.ir)

Version: 4 Date: 13 Dec 2019

Author’s response to reviews:
Dear Editor

The following changes had been made;

1. Abbreviations
Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Abbreviations
ATG: Anti-thymocyte globulin
BMI: Body mass index
CRF: Chronic renal failure
CKD: Chronic kidney disease
eGFR: Estimated Glomerular Filtration Rate
ESRD: End-stage renal disease
LDKT: Living donor kidney transplantation

2. Consent to participate
In your “ethical approval and consent to participate” section, please confirm whether informed consent obtained was written or verbal.
An informed written consent was taken before surgery based on the ethical code of Tehran University of Medical Sciences ethics committee.

3. Availability of data and material
In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.
Information, data, and photos will be provided upon request by Dr. Seyed Mohammad Kazem Aghamir (mkaghamir@tums.ac.ir).

4. Funding
In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. Urology Research Center of Tehran University of Medical Sciences has role in data analysis, and interpretation of data and editing the manuscript.

5. Authors' Contributions

Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship). SMK.A. conception and design of the work, GH.N. acquisition, analysis, and interpretation of data; F.KH., SR. Y. have drafted the work and collecting data A.A. substantively revised manuscript.